Author's response to reviews

Title: Double dermal sinuses, case illustration.

Authors:

Abolhasan Ertiaei Dr (ertiaei@yahoo.com)
Farideh Nejat Dr (nejat@sina.tums.ac.ir)
Mostafa El Khashab Dr (khashab@yahoo.com)

Version: 4 Date: 28 June 2008

Author's response to reviews: see over
Dear Editor

We went through the comments of both referees. We appreciate their efforts to review this manuscript but there is no comment to be answered or changed inside the manuscript. Only Amy Malphrus proposed some language correction before publishing but Shahin Behjati has accepted its according to English language writing as is. We revised the manuscript for English writing and we hope this correction is the same that the reviewer had considered.

Related to the payment for publishing the paper, we will pay the mentioned amount after acceptance of this paper.
We are looking forward to receive your opinions about the current revision of this manuscript.

Regards

Abolhasan Ertiaei M.D

Farideh Nejat M.D, MPH

Mostafa El Khashab MD, PHD